Author's response to reviews

Title: Traumatic Pulmonary Pseuodocysts: Two Case Reports

Authors:

Bulent Kocer (drbkocer@gmail.com)
Gultekin Gulbahar (mdgultekin@gmail.com)
Nesimi Gunal (nesimigunal@yahoo.com.tr)
Koray Dural (koraydural@yahoo.com)
Unal Sakinci (unalsakinci@hotmail.com)

Version: 2 Date: 4 April 2007

Author's response to reviews:

1. The Conclusions section added after the Discussion
2. Competing interests, the individual contributions of the authors and Acknowledgements sections are added to text.

With best regards,